Author’s response to reviews

Title: Pharmacokinetic and pharmacodynamic modelling of oral mitiglinide on glucose lowering in healthy Chinese volunteers

Authors:

Shijia Liu (liushijia2011@163.com)
Peidong Chen (5787411155@qq.com)
Yang Zhao (zhaoyang2015@gmail.com)
Guoliang Dai (3322134585@qq.com)
Bingting Sun (sunbt2011@126.com)
Yao Wang (1554521455@qq.com)
Anwei Ding (dinganwei2011@163.com)
Wenzheng Ju (13951672230@163.com)

Version: 3 Date: 12 Jun 2017

Author’s response to reviews:

From:
Shijia Liu
Department of Clinical Pharmacology
Affiliated Hospital of Nanjing University of Chinese Medicine
Nanjing 210029, China
Tel.: +86-25-8661-7141-50523
Fax: +86-25-8655-5033
E-mail: liushijia2011@163.com

To:
Editorial Office
BMC Pharmacology and Toxicology

Dear Editor:

Thank you very much for your careful review on our manuscript (PHAT-D-16-00190). I would like to appreciate the comments from you. We have revised the manuscript. Thanks for your interest in our work and we look forward to receiving your good news.

Sincerely yours

Shijia Liu

Editor Comments:

1) There are still typos and improvements with the English is still required. We strongly recommend that you use one of the Language Editing Services that are noted below.

[Answer] We have made the improvements to the English language in our manuscript. And the English typos and grammar have been edited throughout by a native English editor.

2) There are still. The figure legends should be in the main text and not with the attached figures. The information for Consent for Publication is incorrect since this section is only required if you present data from individuals. There are typos in the Funding section and the funding sources in the Acknowledgement section should go in the Funding Section. Please look at the Submission Guidelines carefully since it will give you the exact information needed for both formatting as well as what information should go in each section. https://bmcpharmacoltoxicol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

[Answer] The formatting issues have been corrected according to your suggestions.

3) In figure 3 you did not label the y-axis and that is necessary.

[Answer] The y-axis in figure 3 has been labeled.
4) For the legend for Figure 4, please include the abbreviations that are used in the actual figures.

[Answer] The abbreviations have been included in the legend for Figure 4.